Author's response to reviews

Title: Malaria prevention reduces in-hospital mortality among severely ill tuberculosis patients: a three-step intervention in Bissau, Guinea-Bissau

Authors:

Raffaella Colombatti (rcolombatti@gmail.com)
Martina Penazzato (martina.penazzato@gmail.com)
Federica Bassani (f.bassani@microlife.biz)
Cesaltina Silva Vieira (cesaltinavieira1@yahoo.com.br)
Antonia Araujo Lourenco (magalour@hotmail.com)
Fina Vieira (fvieira@yahoo.com.br)
Simone Teso (simone.teso@unipd.it)
Carlo Giaquinto (giaquinto@pediatria.unipd.it)
Fabio Riccardi (fariccardi@hotmail.com)

Version: 7 Date: 28 February 2011

Author's response to reviews:

Dear Sir

The Editors of BMC Infectious Disease,

The formatting change (Major revisions) required by the Editorial Production Team were made to the Manuscript and a modified version of the Manuscript has been uploaded.

References: have been checked and are now in correct numerical order.
Reference 29 has been included in the text
Typography: has been checked again

Thank you

Raffaella Colombatti